Author's response to reviews

Title: Discriminative ability of the generic and condition-specific Child-Oral Impacts on Daily Performances (Child-OIDP) by the Limpopo-Arusha School Health (LASH) Project: A cross-sectional study

Authors:

Hawa S Mbawalla (hawa.mbawalla@student.uib.no)
Matilda Mtaya (matilda.mtaya@student.uib.no)
Joyce R Masalu (jmasalu@muhas.ac.tz)
Pongsri Brudvik (ponsri.brudvik@odont.uib.no)
Anne N Astrom (anne.nordrehaug@cih.uib.no)

Version: 4 Date: 20 May 2011

Author's response to reviews:

Please find the final proofreading stage version of the manuscript. We have made some changes in the manuscript requested by the editors.
1) We have removed all highlighted and tracking from the manuscript
2) removed objectives from the abstract
3) made some requested changes in the reference list

Sincerely
Anne Nordrehaug Åstrøm
Bergen 20.052011